Author's response to reviews

Title: Transient peripheral edema following displaced corneal graft after Descemet Stripping Automated Endothelial Keratoplasty (DSAEK): Case Report

Authors:

George D Kymionis (kymionis@med.uoc.gr)
Dimitra M Portaliou (mimi24279@gmail.com)
Takeshi Ide (teyede@hotmail.com)
Sonia H Yoo (syoo@med.miami.edu)

Version: 3 Date: 11 November 2011

Author's response to reviews: see over
Dear Editor,

We would like to submit the revised case report entitled “Transient peripheral edema following displaced corneal graft after Descemet Stripping Automated Endothelial Keratoplasty (DSAEK): Case Report” that has been corrected according to the comments in your last correspondence.

Below we describe the changes we have made:

**Reviewer 2**

*Discretionary revisions:*

*It might be of interest to include the ECD of the patient's cornea, and especially in the area that cleared up spontaneously.*

**Authors’ Response:**

We agree with the Reviewer but even though we attempted measure the ECD in the area of the graft that cleared spontaneously we failed to obtain satisfactory
images with the confocal microscope in order to include that significant information in the manuscript.

Thank you for the constructive suggestions proposed.

Sincerely

George D. Kymionis MD, PhD